

# 

**Citation:** Cortés V, Cruz A, Onetti S, Kinzel D, Garcia J, Ortiz S, et al. (2021) *Trypanosoma cruzi* infection follow-up in a sylvatic vector of Chagas disease: Comparing early and late stage nymphs. PLoS Negl Trop Dis 15(9): e0009729. https://doi. org/10.1371/journal.pntd.0009729

Editor: João L. Reis-Cunha, Universidade Federal de Minas Gerais, BRAZIL

Received: April 25, 2021

Accepted: August 13, 2021

Published: September 20, 2021

**Copyright:** © 2021 Cortés et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting information files.

**Funding:** This study was funded by the National Research and Development Agency- National Fund for Scientific and Technological Development (https://www.anid.cl/) N°1190392 (AS), the National Research and Development Agency-National Fund for Scientific and Technological Development (https://www.anid.cl/) N°1170367 (CBM), and the National Research and **RESEARCH ARTICLE** 

# *Trypanosoma cruzi* infection follow-up in a sylvatic vector of Chagas disease: Comparing early and late stage nymphs

Valeria Cortés<sup>1</sup>°, Amalia Cruz<sup>1</sup>°, Sofia Onetti<sup>1</sup>, Daniela Kinzel<sup>1</sup>, Javiera Garcia<sup>1</sup>, Sylvia Ortiz<sup>1</sup>, Angélica Lopez<sup>1</sup>, Pedro E. Cattan<sup>2</sup>, Carezza Botto-Mahan<sup>3</sup>\*, Aldo Solari<sup>®</sup>

1 Programa Biología Celular y Molecular, ICBM, Facultad de Medicina, Universidad de Chile, Santiago, Chile, 2 Facultad de Ciencias Veterinarias y Pecuarias, Universidad de Chile, Santiago, Chile, 3 Facultad de Ciencias, Universidad de Chile, Santiago, Chile

These authors contributed equally to this work.

\* cbotto@uchile.cl (CBM); asolari@uchile.cl (AS)

# Abstract

Chagas disease is caused by Trypanosoma cruzi and transmitted by the triatomine Mepraia spinolai in the southwest of South America. Here, we examined the T. cruzi-infection dynamics of field-caught M. spinolai after laboratory feeding, with a follow-up procedure on bug populations collected in winter and spring of 2017 and 2018. Bugs were analyzed twice to evaluate T. cruzi-infection by PCR assays of urine/fecal samples, the first evaluation right after collection and the second 40 days after the first feeding. We detected bugs with: the first sample positive and second negative (+/-), the first sample negative and second positive (-/+), and with both samples positive or negative (+/+; -/-). Bugs that resulted positive on both occasions were the most frequent, with the exception of those collected in winter 2018. Infection rate in spring was higher than winter only in 2018. Early and late stage nymphs presented similar T. cruzi-infection rates except for winter 2017; therefore, all nymphs may contribute to T. cruzi-transmission to humans. Assessment of infection using two samples represents a realistic way to determine the infection a triatomine can harbor. The underlying mechanism may be that some bugs do not excrete parasites unless they are fed and maintained for some time under environmentally controlled conditions before releasing T. cruzi, which persists in the vector hindgut. We suggest that T. cruzi-infection dynamics regarding the three types of positive-PCR results detected by follow-up represent: residual T. cruzi in the rectal lumen (+/-), colonization of parasites attached to the rectal wall (-/+), and presence of both kinds of flagellates in the hindgut of triatomines (+/+). We suggest residual T. cruziinfections are released after feeding, and result 60-90 days after infection persisting in the rectal lumen after a fasting event, a phenomenon that might vary between contrasting seasons and years.

Development Agency-National Fund for Scientific and Technological Development (https://www.anid. cl/) N° 1180940 (PEC). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

## Author summary

In the vector-borne transmission of Chagas disease, approximately 150 species of triatomine bugs are potential vectors for the parasite *Trypanosoma cruzi*. A competent vector must fulfill several features such as the ability to host, amplify, and differentiate the parasite, allowing *T. cruzi* persistence within the insect vector. Our aim was to describe the dynamics of *T. cruzi* infection in a competent triatomine species collected in two contrasting seasons—with different environmental temperatures—of 2017 and 2018. We used a follow-up procedure including *T. cruzi* detection right after collection and 40 days later; both detections were performed after laboratory feeding. Most infected bugs were *T. cruzi* positive on both occasions. However, infected bugs from winter 2018 presented switches from *T. cruzi* negative at collection to *T. cruzi* positive 40 days later. The results suggest infections with *T. cruzi* attached to the hindgut wall as the colonization site, caused by infections that persist there after a fasting event, are released after a second feeding.

# Introduction

Climatic variation is a major determinant of infectious disease dynamics [1]. Insects are ectotherms with a relatively small range of temperatures that can be endured, and in which activity takes place, with variable low thermal limits and narrow upper thermal tolerance [2]. Pathogen transmission in vector-borne diseases can be measured by the basic reproductive number. This epidemiological metric depends on vector and parasite traits including vector competence and density, which can be partitioned as the product of several parameters including development rate, vector survival, feeding, dispersal and reproduction, all dependent on temperature and affecting population abundance [3,4].

Chagas disease, caused by the protozoan parasite *Trypanosoma cruzi*, is the main neglected vector-borne disease in America [5]. This disease presents at least two phases, acute and chronic. The agent, *T. cruzi*, is a stercorarian trypanosome transmitted by hematophagous long-lived insect vectors of the subfamily Triatominae. The infection begins when the fecal/ urine excretion of an infected triatomine contacts the host in the skin wounds, mucous membranes, or by oral ingestion. Thus vector competence depends on several factors including the parasite infection prevalence and the parasite burden in different seasons of the year. As shown in early studies, the highest number of acute cases of Chagas disease occurs during hot seasons; therefore disease occurrence has been associated with temperature changes in different seasons in southern latitudes [6–8], since seasonal temperature variation increases with latitude [9]. Even though some information has been reported on the effect of temperature on domestic triatomine species, little is known about its effect on infected sylvatic triatomine species with a relevant role in *T. cruzi* maintenance and interplay between the domestic and sylvatic transmission cycles of Chagas disease.

*T. cruzi* is a unicellular flagellate trypanosomatid, with a single mitochondrion containing kinetoplast DNA composed of concatenated maxicircles and minicircles [10]. Due to the high number of minicircles, these are appropriate targets for PCR DNA-based detection of *T. cruzi*. Both epimastigotes (non-infective forms) and trypomastigotes attach to the cuticular lining and colonize the rectum [11]. *T. cruzi* competent vectors quickly urinate/defecate several drops after a blood meal. *T. cruzi* attaches to the cuticle of the rectum, covered by a wax layer over the external epicuticle [12]. The parasite amplifies as epimastigotes in the posterior midgut and hindgut, increasing the parasite density there after infection [13–15]. Later on, triatomines release metacyclic trypomastigotes (infective forms) after metacyclogenesis of the

epimastigotes attached to the cuticular lining, another important aspect to determine vector competence [16]. After an extensive blood meal, the triatomine abdominal distension induces a rapid diuresis, and subsequently molting and ecdysis several days later. Parasite density in the hindgut is negatively influenced by fasting [13,17,18], or by regular feeding without reinfection [19–21].

At least three endemic secondary and sylvatic triatomine species are present in the subtropical Pacific side of South America, specifically north-central Chile (~18°S to 34°S), which have arid and semiarid-Mediterranean climatic areas. The triatomine *Mepraia spinolai* [22] mainly inhabits interior valleys up to 3000 m elevation (26°-34°S), where it feeds on mammal species [23–25], with two rodent species (*Phyllotis darwini* and *Octodon degus*) as the most frequent blood sources [24,26]. However, several home as well as peridomiciliary invasion complaints are notified each year [27,28], with human blood as part of the diet [24,25]. Even though vector-borne transmission by the domestic vector *Triatoma infestans* was declared interrupted in Chile in 1999, wild vectors are still an important problem in endemic areas [29,30] and their study should be a priority in public health programs.

*M. spinolai* is distributed in areas with highly variable minimum and maximum temperatures depending on the season of the year, exhibiting an aggregated distribution using rocky outcrops and bromeliads as refuges, where they coexist with small wild mammals [31–33]. This triatomine bug has reduced dispersal capability, with higher activity during the photophase and larger home range in summer than winter [34,35]. In warm seasons, *M. spinolai* takes blood meals on a wider range of hosts compared to the fall season, suggesting that feeding is associated with a greater dispersion pattern and greater host availability [24,35]. Populations of *M. spinolai* collected in warm seasons present highly variable *T. cruzi* infection rates [31,32,36,37], in which the rodent *P. darwini* accounts for most variation in vector infection risk [33].

*M. spinolai* may reach very high densities and exhibits variable population structure depending on the season, with a high proportion of I to III stage nymphs in late austral summer-fall (January to April) and all development stages in austral spring-early summer (September to December) [38–40]. A recent regional study ( $26^{\circ}-31^{\circ}S$ ) showed that mean temperature of the warmest trimester is positively associated with *M. spinolai* abundance and *T. cruzi* infection [32]. A small spatial scale study showed that *T. cruzi* infection in *M. spinolai* is higher in spring and summer compared to fall, and survivorship of the second stage nymphs is lower in spring than in the other seasons [41].

Competent triatomine vectors may harbor *T. cruzi* for long time, an epidemiological trait associated with the risk of transmission to humans, but despite its relevance, there is no reported information about the dynamics of this harboring process in different stage nymphs of sylvatic triatomines captured during contrasting seasons. In this study we used field-caught *M. spinolai* collected during the austral winter and spring seasons of 2017 and 2018 to answer the following questions: (i) Does *T. cruzi* infection prevalence in *M. spinolai* vary between two contrasting seasons of two consecutive years? (ii) Do naturally infected *M. spinolai*, fed in laboratory conditions, provide relevant information to assess variation in *T. cruzi*-infection dynamics? (iii) Does vector development stage (early and late nymphs) associate with *T. cruzi* infection prevalence depending on the follow-up and season?

#### Methods

#### **Ethics statement**

Mice (age 2 months) were obtained from the vivarium facilities of the Faculty of Medicine, University of Chile. All procedures of animal handling carried out in this study were performed according to the rules and with the permission of the Animal Ethics Committee of the University of Chile (CBA#0443-FMUCH-2011; CBA#0987-FMUCH-2018).

#### Triatomine collection and temperature data

The field work was carried out at El Cuyano (31°29'01"S, 71°03'40" W; Coquimbo Region, Chile; Fig 1), a locality included in a hyperendemic area of Chagas disease, during the winter and spring seasons of 2017 and 2018. Minimum and maximum temperatures were obtained from the weather station of Las Chinchillas National Reserve (National Forest Corporation, Ministry of Agriculture of Chile; Supporting information S1 Table. Environmental temperature data), a protected area located ~5.9 km distant from the *M. spinolai* collection site. The study site has a semiarid-Mediterranean climate with widely variable temperatures throughout the year, scarce plant cover dominated by bromeliads, large areas of rock piles, pebbles and sand, and with several domestic (goats, sheep, dogs), native and free-ranging introduced small mammals.



**Fig 1. Study site and environmental temperatures.** Map of North-central Chile showing in hatch pattern the distribution of *Mepraia spinolai* and the study site, El Cuyano (left). Photographs of the rocky outcrop where *M. spinolai* are collected and a late stage nymph of *M. spinolai* (top right). Boxplot of minimum (blue boxes) and maximum (red boxes) temperatures from October 2016 to December 2018, indicating the capturing month of the four groups of *M. spinolai* under study: winter and spring 2017 (W-17, S-17) and 2018 (W-18, S-18) (bottom right). The proportion of collected bugs per development stage is shown for each capture event: early (I and II) and late (III, IV, and V) stage nymphs, and adults (A). Photos by Carezza Botto-Mahan. Base layer of the map available at https://www.ide.cl/index.php/informacion-territorial/descargar-informacion-territorial.

https://doi.org/10.1371/journal.pntd.0009729.g001

## PLOS NEGLECTED TROPICAL DISEASES

Group code	Season/year	Stage	Total	Num	Dead	
				Molted	Non-molted	
W-17	Winter 2017	I-II	42	37	3	2
		III-IV-V	32	16	15	1
		Adults	0	-	-	-
S-17	Spring 2017	I-II	30	19	11	0
		III-IV-V	18	6	11	1
		Adults	9	-	-	0
W-18	Winter 2018	I-II	61	47	11	3
		III-IV-V	76	59	17	0
		Adults	0	-	-	-
S-18	Spring 2018	I-II	60	34	26	0
		III-IV-V	134	51	82	1
		Adults	5	_	-	0

Table 1. Number of Mepraia spinolai individuals, shown by development stage, collected in the field in two seasons of two consecutive years and laboratory fed.

Number of *M. spinolai* molted, non-molted and dead after one feeding event under laboratory conditions. I-II: 1<sup>st</sup> and 2<sup>nd</sup> stage nymphs (early stage nymphs); III-IV-V: 3<sup>rd</sup>, 4<sup>th</sup> and 5<sup>th</sup> stage nymphs, respectively (late stage nymphs).

https://doi.org/10.1371/journal.pntd.0009729.t001

Triatomine bugs were collected manually as bugs approached in an area of  $\sim 1680 \text{ m}^2$  from 11:00 to 16:00 by a trained researcher acting as human bait. Bugs were collected during four and two consecutive days in the winter and spring seasons, respectively. Collected bugs were stored in separate containers with folded paper as refuge and transported in polystyrene foam boxes to the laboratory. Information on each *M. spinolai* collection event is provided in Table 1. A small fraction of nymphs did not resist the trip and arrived dead to the laboratory, mostly starved first-stage nymphs that probably had never fed on a host. These nymphs were not included in the analyses.

# Triatomine fecal sample collection after laboratory feeding

Bugs arrived at the laboratory 1–2 days after collection, where they were classified by development stage and individually housed in a plastic box with several small-size compartments  $(3.2 \times 3.6 \times 1.5 \text{ cm})$  with an identification tag. All bugs were maintained in a growth chamber under optimal conditions at 26 °C, 75% relative humidity and 14:10 (L:D) h cycle [41]. Over a one week period after arrival, bugs were individually fed on uninfected mice (*Mus musculus*) anesthetized with 2% sodium thiopental.

To avoid bug harm and to allow further follow-up evaluation, fecal samples were obtained after spontaneous defecation up to 30 min after feeding, which were mixed with 100  $\mu$ L of distilled water for molecular analyses. In addition, one small volume of each fresh fecal sample was diluted 10 times in saline buffer to be examined microscopically, by searching for *T. cruzi* (free swimming and non-motile flagellates) during 10 min under 100× magnification, following the method described for blood samples [42] adapted for fecal samples. The first feeding nourished the bugs and kept them alive until the second collection of feces/urine samples repeated 40 days later, following the same feeding procedure. Molting and survival were recorded weekly after the first feeding. Dead bugs after the first feeding event were recorded but not included in statistical analyses.

# DNA extraction and detection of T. cruzi infection

Each fecal sample was extracted in a final volume of 200 µL with conditions already described with the EZNA Blood DNA Mini kit (OMEGA BIO-TEK, Norcross, GA, USA). *T. cruzi* 

infection was determined three times by PCR directed to minicircle kDNA with oligos 121 and 122 using different volumes of extracted DNA as template (range of  $2-6 \mu$ L) in a final volume of 50  $\mu$ L [43]. A 10  $\mu$ L volume of each reaction mixture was run in an agarose gel electrophoresis and stained with ethidium bromide. PCR assays were repeated when negative results were obtained. In those cases, the extracted DNA was concentrated three times by evaporation. Three different PCR assays were repeated using variable volumes of the concentrated DNA as template. A negative sample resulted when all PCR attempts with extracted DNA failed to detect a 330 bp amplicon.

## Statistical analyses

We tested for differences in minimum and maximum temperatures between the same austral seasons in the two sampled years by Kruskal-Wallis tests using two months in each season. The association between follow-up/season/development stage and *T. cruzi* prevalence was tested by  $\chi^2$  tests for: (i) follow-up within each season and year, (ii) seasons within year (winter/spring) considering follow-up, and (iii) development stage within each season and year. The significance level (or alpha level) considered statistically significant in this study was 0.05. Analyses were performed with R (version 3.6.0, R Development Core Team 2019) or JMP-Pro (version 14).

# Results

## M. spinolai collection and environmental temperatures

Table 1 shows the distribution of nymphs and adults collected in the austral winter and spring of 2017 (groups W-17 and S-17, hereafter) and 2018 (groups W-18 and S-18, hereafter), shown as number of first and second stage nymphs (early stage nymphs, hereafter), and third, fourth and fifth stage nymphs (late stage nymphs, hereafter). This table also shows the number of bugs that molted after the first feeding and died during the follow-up. Only 14 out of 467 bugs were adults, collected exclusively in spring seasons (S-17 and S-18). Minimum and maximum temperatures revealed statistically significant differences between the studied years (Fig 1). The maximum temperature increased almost 5 °C in winter 2018 compared to winter 2017 (max. median T°: W-17 = 15.6 °C, W-18 = 20.4 °C;  $\chi^2$  = 40.01, p < 0.0001), and 3.4 °C in spring 2018 compared to spring 2017 (max. median T°: S-17 = 24.4 °C, S-18 = 27.8 °C;  $\chi^2$  = 29.65, p < 0.001). The minimum temperature decreased almost 1°C during winter 2018 compared to winter 2017 (min. median T°: W-17 = 4.0 °C, W-18 = 3.2 °C;  $\chi^2$  = 39.75, p < 0.001), and 1.8 °C in spring 2018 compared to spring 2017 (min. median T°: S-17 = 6.8 °C, S-18 = 5.0 °C;  $\chi^2$  = 15.98, p < 0.001). Because of the statistical differences in minimum and maximum temperatures between the same seasons of the years, data of different years were not combined for posterior analyses. Complete environmental temperature data in S1 Table.

# T. cruzi infection dynamics and microscopic observation

Table 2 shows the results of PCR assays of samples obtained with the first fecal/urine sample (i.e., bugs analyzed upon arrival) and those with the second fecal/urine sample (surviving bugs analyzed 40 days after the first laboratory feeding). Three kinds of PCR positive results were recorded: (i) bugs with the first fecal sample positive and the second negative ((+/-), hereafter), (ii) bugs with the first fecal sample negative and the second positive ((-/+), hereafter), and (iii) bugs with both fecal samples positive ((+/+), hereafter) (S1 Fig shows PCR results of DNA bands from representative samples of *T. cruzi* infected and uninfected bugs during the

Group code	Development stage	N° bugs tested	N° bugs (first PCR/second PCR)			Non-infected (%)	Infected (%)		
			(+/-)	(+/+)	(-/+)	(-/-)	(-/-)	First PCR	Follow-up
W-17	All	71	5	20	11	35	49.3	28.2	43.7
	I-II	40	3	4	2	31	77.5	10.0	15.0
	III to V	31	2	16	9	4	12.9	51.6	80.7
S-17	All	56	5	17	6	28	50.0	30.4	41.1
	I-II	30	3	7	1	19	63.3	23.3	26.7
	III to Adults	26	2	10	5	9	34.6	38.5	57.7
W-18	All	134	18	11	25	80	59.7	8.2	26.9
	I-II	58	7	5	14	32	55.2	8.6	32.8
	III to V	76	11	6	11	48	63.2	7.9	22.4
S-18	All	198	41	61	25	71	35.9	30.8	43.4
	I-II	60	11	24	7	18	30.0	40.0	51.7
	III to Adults	138	30	37	18	53	38.4	26.8	39.9

Table 2. PCR results and percentage of *Trypanosoma cruzi* infection in *Mepraia spinolai* collected in winter (W) and spring (S) of 2017 (17) and 2018 (18) surviving two feeding events, shown combined and by early (I-II) and late (III to adults) stages.

First PCR performed upon arrival at the laboratory and the second 40 days later. (+/-): first positive and second negative; (+/+): both positive; (-/+): first negative and second positive; (-/-): both negative. First PCR includes (+/+) only; follow-up includes (+/+) and (-/+). (+/-) probably corresponds to bugs that excreted dead parasites in the first PCR (i.e., with *T. cruzi* DNA remains).

https://doi.org/10.1371/journal.pntd.0009729.t002

follow-up). The number of cases (all development stages combined) of each of the four potential results (+/+, +/-, -/+, -/-) depicted in Table 2 are shown in Fig 2 as percentages for each *M*. *spinolai* group.

Overall, 38.3% of the tested bugs were *T. cruzi* infected in the follow-up, but only 25.1% showed evidence of infection (mostly metacyclic trypomastigotes and few non-motile flagellates) by means of microscopy including results after the two feeding events combined (W-17: 28.2%, S-17: 32.1%, W-18: 14.9%, S-18: 28.8%; Supporting information S1 Table). All groups, with the exception of W-18, exhibited a higher proportion of bugs (all development stages combined) that maintained *T. cruzi* infection (+/+) (W-17:  $\chi^2 = 14.25$ , p < 0.001; S-17:  $\chi^2 = 14.25$ , p < 0.001; S-18:  $\chi^2 = 23.05$ , p < 0.001). In contrast, in W-18 switching from (-) to (+) was the most frequent type of infection ( $\chi^2 = 8.17$ , p = 0.017). No significant differences were detected between seasons for either years in the proportion of bugs (all development stages combined) exhibiting (+/-) results (W-17 vs S-17:  $\chi^2 = 0.19$ , p = 0.665; W-18 vs S-18:  $\chi^2 = 0.019$ , p = 0.890).

# Effect of follow-up on T. cruzi detection

Because +/- results probably correspond to bugs that excreted dead parasites in the first PCR (i.e., with *T. cruzi* DNA remains), they were not considered as *T. cruzi* infected bugs in all posterior statistical analyses. Comparisons of *T. cruzi* prevalence (all development stages combined), for each group and year between the first PCR only or both PCR (follow-up) revealed that the follow-up detected significantly more *T. cruzi* infection in some of the tested groups (Table 2 and Fig 3). Specifically, in winter groups the follow-up detected marginally (W-17:  $\chi^2 = 3.70$ , p = 0.054) and significantly (W-18:  $\chi^2 = 16.13$ , p < 0.001) more infection. The follow-up detected significantly more infection only in spring 2018 (S-18:  $\chi^2 = 6.76$ , p = 0.009). The follow-up did not detect more *T. cruzi* infection in early stage nymphs, except for those nymphs collected in W-2018 ( $\chi^2 = 10.30$ , p = 0.001). In late stage nymphs, the follow-up



Fig 2. Percentage of *Mepraia spinolai* (all development stages combined) with presence of *Trypanosoma cruzi* DNA detected by PCR in fecal/urine samples during two laboratory feedings, in the four collecting events (W-17, S-17, W-18, S-18). The four types of PCR results are shown: both fecal samples positive (+/+), first fecal sample negative and second positive (-/+), first fecal sample positive and second negative (+/-), and both fecal samples negatives (-/-).

https://doi.org/10.1371/journal.pntd.0009729.g002

showed almost the same pattern as for all development stages combined (W-17:  $\chi^2 = 5.83$ , p = 0.016; W-18:  $\chi^2 = 6.20$ , p = 0.013; S-18:  $\chi^2 = 5.28$ , p = 0.022).

#### Effect of season on T. cruzi infection prevalence

Comparison of *T. cruzi* prevalence (all development stages combined) between seasons of the same year, with the first PCR only or both PCR (follow-up), revealed significant differences for



Fig 3. Percentage of *Trypanosoma cruzi* infection (all development stages combined) in each *Mepraia spinolai* group (W-17, S-17, W-18, S-18) for the first PCR (+/+) only or follow-up (+/+ and -/+). P-values are shown above each comparison. Statistically significant differences are in bold. (+/-) probably corresponds to bugs that excreted dead parasites in the first PCR (i.e., with *T. cruzi* DNA remains).

https://doi.org/10.1371/journal.pntd.0009729.g003

2018 only. Specifically, in spring 2018 we detected more *T. cruzi* infection prevalence than winter, considering both the first PCR only ( $\chi^2 = 24.03$ , p < 0.001) and the follow-up ( $\chi^2 = 9.44$ , p = 0.002). The same pattern as for all development stages combined was detected in early and late stage nymphs in 2018. Conversely, in 2017 we detected marginally significant difference between seasons only in late stage nymphs with the follow-up procedure ( $\chi^2 = 3.56$ , p = 0.059).

#### Effect of development stage on T. cruzi infection prevalence

*T. cruzi* prevalence between early and late stage nymphs within the same season and year, considering the first PCR only, revealed differences in only one group. Specifically, higher *T. cruzi* prevalence was detected in W-17 in late stage nymphs compared to early ones ( $\chi^2 = 14.95$ , p < 0.001). No statistically significant differences were detected in the other groups (S-17, W-18, and S-18).

## Discussion

We performed a *T. cruzi* follow-up in naturally infected bugs subjected to laboratory feedings, one feeding right after collection and a second 40 days later. Overall, in spring we detected more *M. spinolai* individuals infected than in winter, both right after collection and after feeding in the lab (follow-up). The same pattern was reported by a previous study carried out in one location nearby our study site [31], in which summer infection was higher than that detected in winter bugs right after capture. However, unlike previous studies, the follow-up procedure would allow to find out the type of infection a bug is experiencing when captured under field conditions, rendering valuable information on vector infectivity.

In our study, the most frequent infection corresponded to bugs that excreted *T. cruzi* after the first and the second feeding (+/+), which represents an active infection, i.e., bugs with parasites in the rectal lumen as well as attached to the hindgut wall. Other individuals switched from positive to negative PCR results (+/-), which could represent a residual infection with *T. cruzi* only in the rectal lumen of starved insects, probably occurring 60–90 days after infection, with dead detached parasites as previously reported [17]. However, other factors such as the insect immune response might explain this result. The infection (-/+) could represent insects with *T. cruzi* attached to the hindgut. This situation probably occurs in more recent infections, when parasites attached to the hindgut are released after feeding and the excreta analyzed 40 days later or less [17]. Nonetheless, only insect dissection might allow to evaluate where *T. cruzi* is developing.

Early and late stage nymphs (including adults) from both seasons and years were analyzed all together and separately to gather information on development stage-dependent *T. cruzi* prevalence and infection risk, especially because nymphs of different development stages might present contrasting eco-physiological traits (e.g., dispersal capability, temperature endurance) [44]. Higher *T. cruzi* prevalence with the follow-up procedure was detected in late stage nymphs in three out of the four studied groups (except for spring 2017) compared to the prevalence detected with the first PCR only. In early stage nymphs the follow-up procedure was able to detect higher *T. cruzi* prevalence only in winter 2018. This last result probably corresponds to newly infected early stage nymphs, after feeding on infected mammals in an unusual warmer winter such as that of 2018, especially if we consider that *M. spinolai* is active over  $15 \degree C$  [34,35]. In summary, we suggest that the results of the first PCR would be showing vector infectivity (at the time of capture), but the follow-up would be detecting also ongoing infection within a vector [21].

T. cruzi prevalence was higher in early stage nymphs in spring than in winter of 2018 only, using both detection procedures. In general, late stage nymphs from both spring seasons presented higher T. cruzi prevalence than those from winter seasons. We suggest that minimum and maximum environmental temperatures perceived by the winter and spring 2018 groups were more contrasting than those of 2017 groups. Even though we detected some interannual variation, we suggest that environmental factors translated into higher T. cruzi prevalence in the warmer season of 2018, indicating that season could be a relevant factor when assessing infection risk in austral latitudes. Metacyclogenesis is a temperature-dependent process [45]. For example, at 22–23 °C fewer metacyclic trypomastigotes are excreted by triatomines, but many more are released at 26-30 °C [45,46]. In other study, trypomastigotes appeared in the rectum and feces earlier at 28° than at 20°C, but after that time triatomines developed similar population densities [47], probably reaching their carrying capacities. We found mostly metacyclic trypomastigote forms by optical microscopy in the excreta of highly infected *M. spinolai* in spring, supporting that T. cruzi infective forms are released with high infectivity to different hosts mostly in warmer seasons. The sylvatic species Triatoma brasiliensis presented higher T. cruzi infection rates in equator latitudes of Brazil in the rainy season than in the dry season [48]. These results indicate that in areas with subtle changes in environmental temperatures, other variables such as rainfall rather than temperature may determine vector infectivity.

Late stage nymphs resulted in higher *T. cruzi* infections determined with the first PCR assay than early stage nymphs collected in winter 2017, but not in equivalent bugs of winter 2018. Late stage nymphs of winter 2017 were probably more resistant to low temperatures and may have dispersed larger distances to search for a host than early stage nymphs, having more opportunities to survive and become infected by *T. cruzi*.

The different infection detected between early and late stage nymphs of the two studied winters suggests that subtle changes in environmental threshold temperatures might determine when *M. spinolai* enter fasting, therefore reducing the probability of infection or re-infection. Contrarily, in both springs, similar *T. cruzi* prevalence was detected in early and late stage nymphs. In consequence, both types of nymph stages might be equally risky for mammals including humans in transmitting *T. cruzi* in warmer seasons and, probably, in unusually warm winters.

During the follow-up we detected +/- residual infections for bugs that excrete the last fraction of probably dead parasites from the rectal lumen after a long fasting event. We found that this kind of result was not associated with the studied seasons, because not only winter bugs presented residual infection (+/-), but also spring bugs, suggesting that fasting also occurs in nature in temperate weather.

Different extensions of fasting up to starvation affect the nutritional status of triatomines altering host detection and approaching behavior [49]. A field study detected that *T. cruzi*-infected *M. spinolai* with the highest nutritional status approached humans first [40]. Those results along with the present study in *M. spinolai* of different seasons are epidemiologically important to determine the *T. cruzi* infectivity risk for humans in warm seasons. The two kinds of residual infection detected with a follow-up have a timing not yet determined. However, in *T. infestans* experimental infection with *T. cruzi* (strain Chile 5) under fasting, an old infection is close to 30 days and a much older infection can last 60 or more days with an extensive killing of flagellates in the rectum [17,50]. A long fast in *M. spinolai* during winter can last 60–90 days, probably without molting, a period of time this species can withstand. Under optimal laboratory conditions, uninfected *M. spinolai* [18]. *M. spinolai* of different development stages submitted to fasting after full engorgement had life spans of 70–300 days [41], suggesting that *M. spinolai* can withstand long periods of fasting like those it undergoes in winter.

In conclusion, all development stages may contribute equally to *T. cruzi* transmission to humans mainly, but not exclusively, in warmer seasons. Future studies, using more sensitive detection methods, should focus on parasitic burden follow-up in sylvatic triatomines and the interannual variation in *T. cruzi* prevalence to assess the complete scenario of transmission in austral latitudes of South America.

# **Supporting information**

**S1 Table.** Microscopy observation and PCR results for *M. spinolai*, and environmental temperature data. Detailed information for each *M. spinolai* individual indicating season and year, development stage, first and second optical microscopy and PCR results, and molting. Minimum and maximum daily temperatures from October 2016 to December 2018 from the National Forest Corporation (CONAF) weather station located at Las Chinchillas National Reserve, Coquimbo Region, Chile. (XLSX)

**S1 Fig.** Representative results of PCR assays on fecal/urine samples of some *M. spinolai*. (TIFF)

# Acknowledgments

We specially thank Nora Peña for her invaluable work collecting *M. spinolai* in the field, Esteban San Juan for compiling the map and figures. We are very grateful to Jorge Silva (National Forest Corporation, CONAF) for providing climatic data from Las Chinchillas National Reserve, and José Fernández from the Faculty of Medicine, University of Chile, for providing bibliographic resources.

# **Author Contributions**

Conceptualization: Aldo Solari.

Data curation: Carezza Botto-Mahan, Aldo Solari.

Formal analysis: Carezza Botto-Mahan, Aldo Solari.

Funding acquisition: Aldo Solari.

**Investigation:** Valeria Cortés, Amalia Cruz, Sofia Onetti, Daniela Kinzel, Javiera Garcia, Sylvia Ortiz, Angélica Lopez.

Project administration: Aldo Solari.

Supervision: Aldo Solari.

Validation: Carezza Botto-Mahan, Aldo Solari.

Visualization: Carezza Botto-Mahan.

Writing - original draft: Carezza Botto-Mahan, Aldo Solari.

Writing - review & editing: Pedro E. Cattan, Carezza Botto-Mahan, Aldo Solari.

#### References

 Poulin R. Global warming and temperature-mediated increases in cercarial emergence in trematode parasites. Parasitology. 2006; 132:143–151. https://doi.org/10.1017/S0031182005008693 PMID: 16393363

- 2. Hoffmann AA, Chown SL, Clusella-Trullas S. Responses to global climate change. Upper thermal limits in terrestrial ectotherms: how constrained are they? Funct Ecol. 2013; 27:934–949.
- Sternberg ED, Thomas MB. Local adaptation to temperature and the implications for vector-borne diseases. Trends Parasitol. 2014; 30:115–122. https://doi.org/10.1016/j.pt.2013.12.010 PMID: 24513566
- 4. Fresquet N, Lazzari CR. Response to heat in *Rhodnius prolixus*: the role of the thermal background. J Insect Physiol. 2011; 57:1446–1449. https://doi.org/10.1016/j.jinsphys.2011.07.012 PMID: 21806990
- 5. WHO. World Health Organization: A Global Brief on Vector-Borne Disease. WHO, Geneva; 2014.
- 6. Talice RV, Costa RS, Rial B, Osimani JJ. Los 100 primeros casos agudos confirmados de Enfermedad de Chagas en el Uruguay. Montevideo: Editores A. Monteverde y Cia; 1940.
- Dias E. Informações acerca de 300 casos de doença de Chagas com período inicial conhecido, fichados no Centro de Estudos de Bambuí. E Dias—O Hospital, Hospital (Rio de Janeiro) 1945; 47:647– 653.
- Romaña C. Enfermedad de Chagas. Buenos Aires: López Libreros Editores S.R.L.; 1963. PMID: 13983242
- 9. Sunday JM, Bates AE, Dulvy NK. Global analysis of thermal tolerance and latitude in ectotherms. Proc Biol Sci. 2011; 278:1823–1830. https://doi.org/10.1098/rspb.2010.1295 PMID: 21106582
- Junqueira AC, Degrave W, Brandão A. Minicircle organization and diversity in *Trypanosoma cruzi* populations. Trends Parasitol. 2005; 21:270–272. <u>https://doi.org/10.1016/j.pt.2005.04.001</u> PMID: 15922247
- Böker CA, Schaub GA. Scanning electron microscopic studies of *Trypanosoma cruzi* in the rectum of its vector *Triatoma infestans*. Z Parasitenkd. 1984; 70:459–469. https://doi.org/10.1007/BF00926686 PMID: 6382848
- 12. Zeledón R, Alvarenga RJ, Schosinsky K. Ecology of Trypanosoma cruzi in the insect vector. 1977a; Inter Symp Chagas Disease, New York, PAHO Scientific Publication n°347:59–70
- 13. Schaub GA, Böker CA. Colonization of the rectum of *Triatoma infestans* by *Trypanosoma cruzi*: influence of starvation studied by scanning electron microscopy. Acta Trop. 1986; 43:349–354. PMID: 2882662
- Schaub GA. *Trypanosoma cruzi*: quantitative studies of development of two strains in small intestine and rectum of the vector *Triatoma infestans*. Exp Parasitol. 1989; 68:260–273. <u>https://doi.org/10.1016/</u> 0014-4894(89)90108-2 PMID: 2649388
- 15. de Almeida DF, Guerra B, Rezende VL, Diego PH, Paiva GAC, Vionette do ARJ, et al. Monitoring of the parasite load in the digestive tract of *Rhodnius prolixus* by combined qPCR analysis and imaging techniques provides new insights into the trypanosome life cycle. PLoS Negl Trop Dis. 2015; 9:e0004186. https://doi.org/10.1371/journal.pntd.0004186 PMID: 26496442
- Schaub GA, Lösch P. *Trypanosoma cruzi*: origin of metacyclic trypomastigotes in the urine of the vector *Triatoma infestans*. Exp Parasitol. 1988; 65:174–186. https://doi.org/10.1016/0014-4894(88)90121-x PMID: 3280333
- Kollien AH, Schaub GA. The development of *Trypanosoma cruzi* in Triatominae. Parasitol Today 2000; 16:381–387. https://doi.org/10.1016/s0169-4758(00)01724-5 PMID: 10951597
- Garcia V, Graterol J, López A, Ortiz S, Solari A. Influence of *Trypanosoma cruzi* (Kinetoplastida: Trypanosomatidae) infection on mortality of the sylvatic Triatomine vector, *Mepraia spinolai* (Heteroptera: Reduviidae), under fasting. J Med Entomol. 2019; 56:1384–1388. https://doi.org/10.1093/jme/tjz124 PMID: 31322659
- Dias E. Estudos sobre o Schizotrypanum cruzi. Memórias do Instituto Oswaldo Cruz XXVIII, 1934. Tèse de doutoramento. Faculdade de Medicina da Universidade do Rio de Janeiro.
- Vargas LG, Zeledón R. Effect of fasting on *Trypanosoma cruzi* infection in *Triatoma dimidiata* (Hemiptera: Reduviidae). J Med Entomol. 1985; 22:683. <u>https://doi.org/10.1093/jmedent/22.6.683</u> PMID: 3908682
- Egaña C, Pinto R, Vergara F, Ortiz S, Campos R, Solari A. Fluctuations in *Trypanosoma cruzi* discrete typing unit composition in two naturally infected triatomines: *Mepraia gajardoi* and *M. spinolai* after laboratory feeding. Acta Trop. 2016; 160:9–14. <u>https://doi.org/10.1016/j.actatropica.2016.04.008</u> PMID: 27109041
- 22. Lent Wygodzinsky PW. Revision of the Triatominae (Hemiptera, Reduviidae), and their significance as vectors of Chagas' disease. B Am Mus Nat Hist. 1979; 163:125–520.
- Schenone H, Contreras MC, Borgoño JM, Rojas A, Villarroel F, Valdés J. Chagas' disease in Chile. Rural and periurban sectors of the endemo-enzootic area. Relationship between housing conditions, domiciliary triatomid infestation and infection by *Trypanosoma cruzi* of the vector, humans and domestic mammals. 1982–1985. Bol Chil Parasitol. 1985; 40:58–67. PMID: 3939408

- Canals M, Cruzat L, Molina MC, Ferreira A, Cattan PE. Blood host sources of *Mepraia spinolai* (Heteroptera: Reduviidae), wild vector of Chagas disease in Chile. J Med Entomol. 2001; 38:303–307. https://doi.org/10.1603/0022-2585-38.2.303 PMID: 11296839
- Molina MC, Cattan P, Canals M, Cruzat L, Aguillón JC, Ferreira A. A simple immunometric assay to assess the feeding habits of *Mepraia spinolai*, a *Trypanosoma cruzi* vector. Parasitol Res. 2004; 92:375–379. https://doi.org/10.1007/s00436-003-1011-6 PMID: 14745546
- Chacón F, Bacigalupo A, Quiroga JF, Ferreira A, Cattan PE, Ramírez-Toloza G. Feeding profile of Mepraia spinolai, a sylvatic vector of Chagas disease in Chile. Acta Trop. 2016; 162:171–173. <u>https://doi.org/10.1016/j.actatropica.2016.06.027</u> PMID: 27349188
- Schenone H, Villarroel F, Rojas A, Alfaro E. Biological and ecological factors in the epidemiology of Chagas' disease in Chile. Bol Chil Parasitol. 1980; 35:42–54. PMID: 6797447
- Garrido R, Bacigalupo A, Peña-Gómez F, Bustamante RO, Cattan PE, Gorla DE, et al. Potential impact of climate change on the geographical distribution of two wild vectors of Chagas disease in Chile: *Mepraia spinolai* and *Mepraia gajardoi*. Parasit Vectors. 2019; 12:478. <u>https://doi.org/10.1186/s13071-019-3744-9 PMID: 31610815</u>
- 29. Canals M, González C, Canals L, Canals A, Cáceres D, Alvarado S, Cattan P, Saavedra M, Zulantay I, Apt W. What do the numbers tell us about the temporal evolution of Chagas' disease? Rev Chile Infectol. 2017; 34:120–127. https://doi.org/10.4067/S0716-10182017000200004 PMID: 28632825
- Salas P. Epidemiology of Chagas disease: High mortality and incidence rate, Coquimbo Region. Rev Chile Infectol. 2020; 37:402–412.
- Ihle-Soto C, Costoya E, Correa JP, Bacigalupo A, Cornejo-Villar B, Estadella V, Solari A, Ortiz S, Hernández HJ, Botto-Mahan C, Gorla DE, Cattan PE. Spatio-temporal characterization of *Trypanosoma cruzi* infection and discrete typing units infecting hosts and vectors from non-domestic foci of Chile. PLoS Negl Trop Dis. 2019; 13:e0007170. <u>https://doi.org/10.1371/journal.pntd.0007170</u> PMID: 30768613
- San Juan E, Araya-Donoso R, Sandoval-Rodríguez A, Yáñez-Meza A, Quiroga N, Botto-Mahan C. Lizards and rabbits may increase Chagas infection risk in the Mediterranean-type ecosystem of South America. Sci Rep. 2020; 10:1853. https://doi.org/10.1038/s41598-020-59054-8 PMID: 32024939
- Botto-Mahan C, Bacigalupo A, Correa JP, Fontúrbel FE, Cattan PE, Solari A. Prevalence, infected density or individual probability of infection? Assessing vector infection risk in the wild transmission of Chagas disease: Proc Biol Sci. 2020; 287:20193018. <u>https://doi.org/10.1098/rspb.2019.3018</u> PMID: 32156212
- Canals M, Solis R, Valderas J, Ehrenfeld M, Cattan PE. Preliminary studies on temperature selection and activity cycles of *Triatoma infestans* and *T. spinolai* (Heteroptera: Reduviidae), Chilean vectors of Chagas' disease. J Med Entomol. 1997; 34:11–17. https://doi.org/10.1093/jmedent/34.1.11 PMID: 9086704
- 35. Botto-Mahan C, Cattan PE, Canals M, Acuña M. Seasonal variation in the home range and host availability of the blood-sucking insect *Mepraia spinolai* in wild environment. Acta Trop. 2005; 95:160–163 https://doi.org/10.1016/j.actatropica.2005.05.001 PMID: 15949784
- 36. Coronado X, Rozas M, Botto-Mahan C, Ortíz S, Cattan PE, Solari A. Molecular epidemiology of Chagas disease in the wild transmission cycle: the evaluation in the sylvatic vector *Mepraia spinolai* from an endemic area of Chile. Am J Trop Med Hyg. 2009; 81:656–659. https://doi.org/10.4269/ajtmh.2009.09-0053 PMID: 19815882
- Correa JP, Bacigalupo A, Fontúrbel F, Oda E, Cattan PE, Solari A, Botto-Mahan C. Spatial distribution of an infectious disease in a native small mammal community. Sci Nat. 2015; 102:51. <u>https://doi.org/10.1007/s00114-015-1304-5</u>
- Knierim F, Castro M, Villarroel F, Schenone H. Estudio preliminar sobre la fuente de alimentación de *Triatoma infestans* y *Triatoma spinolai* mediante la reacción de doble difusión en gel. Bol Chil Parasitol. 1976; 31:34–36. PMID: 822861
- **39.** Canals M, Ehrenfeld M, Solís R, Cruzat L, Pinochet A, Tapia C, et al. Comparative biology of *Mepraia spinolai* in laboratory and field conditions: five years study. Parasitol Dia. 1998; 22:72–78.
- 40. Estay-Olea D, Correa JP, de Bona S, Bacigalupo A, Quiroga N, San Juan E, et al. *Trypanosoma cruzi* could affect wild triatomine approaching behaviour to humans by altering vector nutritional status: A field test. Acta Trop. 2020; 210:105574 <u>https://doi.org/10.1016/j.actatropica.2020.105574</u> PMID: 32504588
- McCabe A, Yañez F, Pinto R, López A, Ortiz S, Muñoz-San Martin, et al. Survivorship of wild caught Mepraia spinolai nymphs: The effect of seasonality and *Trypanosoma cruzi* infection after feeding and fasting in the laboratory. Infect Genet Evol. 2019; 71:197–204. https://doi.org/10.1016/j.meegid.2019. 04.002 PMID: 30953715

- Feilij H, Muller L, Gonzalez Cappa SM. Direct micromethod for diagnosis of acute and congenital Chagas' disease. J Clin Microbiol. 1983; 18:327–330. <u>https://doi.org/10.1128/jcm.18.2.327-330.1983</u>
  PMID: 6413530
- 43. Wincker P, Britto C, Pereira JB, Cardoso MA, Oelemann O, Morel CM. Use of a simplified polymerase chain reaction procedure to detect *Trypanosoma cruzi* in blood samples from chronic chagasic patients in a rural endemic area. Am J Trop Med Hyg. 1994; 51:771–777. <u>https://doi.org/10.4269/ajtmh.1994</u>. 51.771 PMID: 7810810
- 44. Guarneri AA, Lazzari C, Xavier AAP, Diotaiuti L, Lorenzo M. The effect of temperature on the behaviour and development of *Triatoma brasiliensis*. Physiol Entomol. 2003; 28:185–191.
- 45. Tamayo LD, Guhl F, Vallejo GA, Ramírez JD. The effect of temperature increase on the development of *Rhodnius prolixus* and the course of *Trypanosoma cruzi* metacyclogenesis PLoS Negl Trop Dis. 2018; 12(8):e0006735. https://doi.org/10.1371/journal.pntd.0006735 PMID: 30110329
- 46. Wood SF. Environmental temperature as a factor in development of *Trypanosoma cruzi* in *Triatoma protracta*. Exp Parasitol. 1954; 3(3):227–233. https://doi.org/10.1016/0014-4894(54)90021-1 PMID: 13161964
- Asin S, Catala S. Development of *Trypanosoma cruzi* in *Triatoma infestans*: influence of temperature and blood consumption. J Parasitol. 1995; 81(1):1–7. PMID: 7876960
- Sarquis O, Carvalho-Costa FA, Oliveira LS, Duarte R, PS DA, de Oliveira TG, et al. Ecology of *Triatoma brasiliensis* in northeastern Brazil: seasonal distribution, feeding resources, and *Trypanosoma cruzi* infection in a sylvatic population. J Vector Ecol. 2010; 35(2):385–394. <u>https://doi.org/10.1111/j.1948-7134.2010.00097.x</u> PMID: 21175946
- **49.** Schofield CJ. Nutritional status of domestic populations of *Triatoma infestans*. Trans R Soc Trop Med Hyg. 1980; 74(6):770–778. https://doi.org/10.1016/0035-9203(80)90197-2 PMID: 7010698
- **50.** Kollien AH, Schaub GA Development of *Trypanosoma cruzi* after starvation and feeding of the vector a review. Tokai J Exp Clin Med. 1999; 23:335–340.